# Lagged Wetland CH<sub>4</sub> Flux Response in a Historically Wet Year

# J. Turner<sup>1</sup>, A. R. Desai<sup>2</sup>, J. Thom<sup>3</sup>, and K. P. Wickland<sup>4</sup>

<sup>1</sup>Freshwater and Marine Sciences, University of Wisconsin-Madison, Madison, WI, USA

<sup>2</sup>Department of Atmospheric & Oceanic Sciences, University of Wisconsin-Madison, Madison, WI, USA

<sup>3</sup>Space Science and Engineering Center, University of Wisconsin-Madison, Madison, WI, USA <sup>4</sup>United States Geological Survey, Water Resources Mission Area, Boulder, CO, USA

Corresponding author: J. Turner (jlturner4@wisc.edu)

# **Key Points:**

- Analyzed lagged response of methane flux to different driver variables at two closely located fen wetlands in Wisconsin
- Air-temperature normalization of methane flux was crucial for interpretation of lagged responses, especially in wet year
- Lagged response of methane flux to gross primary productivity surpassed sixty days and had weaker correlation during wet year at both sites

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2021JG006458.

#### Abstract

While a stimulating effect of plant primary productivity on soil carbon dioxide  $(CO_2)$ emissions has been well documented, links between gross primary productivity (GPP) and wetland methane (CH<sub>4</sub>) emissions are less well investigated. Determination of the influence of primary productivity on wetland CH<sub>4</sub> emissions (FCH<sub>4</sub>) is complicated by confounding influences of water table level and temperature on CH<sub>4</sub> production, which also vary seasonally. Here, we evaluate the link between preceding GPP and subsequent FCH<sub>4</sub> at two fens in Wisconsin using eddy covariance flux towers, Lost Creek (US-Los) and Allequash Creek (US-ALQ). Both wetlands are mosaics of forested and shrub wetlands, with US-Los being larger in scale and having a more open canopy. Co-located sites with multi-year observations of flux, hydrology, and meteorology provide an opportunity to measure and compare lag effects on FCH<sub>4</sub> without interference due to differing climate. Daily average FCH<sub>4</sub> from US-Los reached a maximum of 47.7  $\eta$ mol CH<sub>4</sub> m<sup>-2</sup>·s<sup>-1</sup> during the study period, while US-ALQ was more than double at 117.9 nmol CH<sub>4</sub> m<sup>-2</sup>·s<sup>-1</sup>. The lagged influence of GPP on temperature-normalized FCH<sub>4</sub> (T<sub>air</sub>-FCH<sub>4</sub>) was weaker and more delayed in a year with anomalously high precipitation than a following drier year at both sites. FCH<sub>4</sub> at US-ALQ was lower coincident with higher stream discharge in the wet year (2019), potentially due to soil gas flushing during high precipitation events and lower water temperatures. Better understanding of the lagged influence of GPP on FCH<sub>4</sub> due to this study has implications for climate modeling and more accurate carbon budgeting.

#### **Plain Language Summary**

Research on what controls wetland methane emissions is continually advancing, and while this is beneficial for predicting future climate scenarios, there is still a need to understand how changes in plant productivity will influence wetland methane emissions. In this study, we investigated the strength and lag time of the relationship between gross primary productivity due to photosynthesizing plants and wetland methane flux in two closely situated sites. We also looked at how hydrology might change that relationship. We found the total amount of methane emitted in an extremely wet year was less than what was emitted in the following drier year at both wetlands potentially because of less carbon provided to the soil by photosynthesizing plants. The difference in methane emissions from one year to the next could be influenced by wetland

hydrology, water temperature, or other conditions that impact methane-producing bacteria. Results from this study will help scientists better predict methane emissions following high precipitation years which may become more common in a changing climate.

# Introduction

1

By the year 2100, mean global annual CH<sub>4</sub> flux (FCH<sub>4</sub>) from natural wetlands is projected to increase from 172 Tg CH<sub>4</sub> yr<sup>-1</sup> to anywhere between 222 and 338 Tg CH<sub>4</sub> yr<sup>-1</sup> depending on the climate scenario (Zhang et al., 2017). Under the best climate scenario of strong climate mitigation (RCP 2.6), wetland methane ( $CH_4$ ) emissions are projected to decline in the 2050s after peaking at ~225 Tg CH<sub>4</sub> yr<sup>-1</sup>. Radiative forcing feedback from wetland CH<sub>4</sub> could account for a large portion of the total radiative forcing change from CH<sub>4</sub>, accounting for  $0.04 \pm$  $0.002 \text{ Wm}^{-2}$ , and global mean temperature would increase slightly as a result (Zhang et al., 2017). Ecosystem-scale controls over microbial activity and resulting wetland CH<sub>4</sub> emissions are difficult to include in climate projection models despite their importance as a major climate feedback. Specifically, there is a need to understand and include the impact of shifting spatial patterns of vascular plants on CH<sub>4</sub> transport from soil into atmosphere, and biogeographical distribution of methanogen communities and their metabolic processes, which could be leading to current underestimation of CH<sub>4</sub> emissions with certain models. A better understanding of the relationship between gross primary productivity (GPP) and FCH<sub>4</sub> in fen wetlands is crucial to understanding the potential impacts of a longer growing season, higher GPP, and the shifting distribution of terrestrial ecosystems due to a changing climate (Zhang et al., 2017). In this study, we take a close look at the lagged effect of GPP on FCH<sub>4</sub> from two closely located north temperate fen wetlands.

Global syntheses of eddy covariance flux data and improved earth system models have contributed to a better understanding of FCH<sub>4</sub> drivers and variability across sites (Knox et al., 2019; Delwiche et al., 2021; Knox et al., 2021), but there is room to improve understanding even further through regional site comparisons. Driver analysis on this small-scale has the potential to explain variability in FCH<sub>4</sub> in locations undergoing the same synoptic meteorology on the scale of hundreds to thousands of kilometers (e.g., low pressure systems), some overlapping mesoscale meteorology at the scale of a few to hundreds of kilometers (e.g., thunderstorms), but separate

microclimates under the scale of a kilometer (e.g., structure and function of vegetation and its influence on local climate variables).

Some prior studies have found no significant relationship between GPP and FCH<sub>4</sub> (Sturtevant and Oechel, 2013; Davidson et al., 2016). However, certain physical mechanisms should cause GPP and FCH<sub>4</sub> to be linked in wetland ecosystems, either synchronously or lagged. GPP can influence FCH<sub>4</sub> directly through plant-mediated transport of gas from porewater to the atmosphere (Dannenberg and Conrad, 1999; Dorodnikov et al., 2011) or indirectly through plant C fixation to soil methanogens during photosynthesis (Hatala et al., 2012). Aerenchymatous wetland plants transport dissolved CH<sub>4</sub> from porewater, through roots, into the root cortex, and then out through leaf sheath micropores in the lower part of the shoot (Nouchi et al., 1990; Henneberg et al., 2012). Root area is therefore an important determinant of plant-mediated CH<sub>4</sub> transport and will increase CH<sub>4</sub> production in anoxic conditions. Ecosystem-scale FCH<sub>4</sub> is more difficult to predict given that flux varies among plants of the same genus (Ding et al., 2005). CH<sub>4</sub> oxidation rate will also peak at different times of the season depending on plant type (Welsch and Yavitt, 2007).

The lag time between plant C assimilation and soil CO<sub>2</sub> efflux (i.e., microbial decomposition & root respiration) takes less than one day for grasses and up to 5 days for mature trees (Kuzyakov and Gavrichkova, 2010). Due to plant metabolism, one would expect, in anoxic soil conditions, a similar lagged influence of GPP on soil FCH<sub>4</sub>, which has been detected and discussed in some studies (Mitra et al., 2020; Bridgham et al., 2013; Updegraff et al., 2001) but not others (Villa et al., 2019) or may disappear after temperature-normalization (Rinne et al., 2018; Chen et al., 2020). Methane emission can be stimulated by plant shoot clipping (which results in the growth of new roots) in as few as three days, although the short duration of mesocosm experiments limits measurement of maximum total lag time (Rietl et al., 2017). Another experiment found a six-day lag between soaking a rice field and a rise in CH<sub>4</sub> emissions and a clear change in the magnitude of the FCH<sub>4</sub> diel cycle depending on plant growth phase (Centeno et al., 2017).

In this study we compare FCH<sub>4</sub> and related environmental variables of two co-located fen wetlands in Wisconsin to answer two questions: (1) What is the influence of plant C fixation on FCH<sub>4</sub> as measured by the lagged effect of GPP at two north temperate fen wetlands when

removing the known influence of air temperature  $(T_{air})$ ? (2) How do factors relating to wetland hydrology as indicated by wetland stream discharge, stream temperature ( $T_{stream}$ ), and water table depth (WTD), mediate the GPP-FCH<sub>4</sub> relationship at both sites? We hypothesize GPP will have a strong but short-term lagged influence on temperature-normalized FCH<sub>4</sub> ( $T_{air}$ -FCH<sub>4</sub>) at both sites because of allocation of recently fixed C to roots, followed by methanogenesis. Removing the influence of  $T_{air}$  will be critical to the interpretation of results. We expect wetland stream discharge to correspond with increasing FCH<sub>4</sub>, assuming it is indicative of a higher WTD. Finally, FCH<sub>4</sub> should be similar but not identical at the two sites in this study given that they are co-located fens with mixed vegetative cover but possess unique physical and hydrological features, discussed below.

## 2 Methods

# 2.1 Site descriptions

Our study focuses on two sites in northern Wisconsin that are located approximately 29 km apart: US-Los and US-ALQ. US-Los (46.082777, -89.978611) is larger in scale (flux footprint radius 1,033 m), features more open canopy vegetation, and is dominated by broad-leaved deciduous shrub vegetation (20% of flux footprint). US-ALQ (46.030759, -89.606730) is smaller in scale (flux footprint radius 238 m), features more sheltered canopies, and is dominated by broad-leaved deciduous or evergreen shrub vegetation (30% of flux footprint). Both sites are mixed sedge meadow, forest, and shrub wetland. Both sites are fen wetlands, as they are surface water and groundwater sourced and have peat soil, and each is bisected by a headwater stream (Lost Creek at US-Los; Allequash Creek at US-ALQ). Detailed site descriptions are available for US-Los in Sulman et al. (2009) and for US-ALQ in Anderson and Lowry (2007). It should be noted that Allequash Creek (flowing through US-ALQ) is a groundwater-fed stream (Pint et al., 2003) and thus flows year-round.

# 2.2 Flux data

CH<sub>4</sub> and CO<sub>2</sub> eddy covariance flux data for US-ALQ (doi: 10.3389/fenvs.2019.00179) and US-Los (doi:10.17190/AMF/1246071) are available on Ameriflux (Olson, B. 2020; Desai, A. 2020; https://ameriflux.lbl.gov/).). All data analyzed in this study were collected during January 1, 2019 through Dec 31, 2020. Instrumentation at both sites included a sonic anemometer (Campbell Scientific, Inc., Logan, 188 UT, CSAT-3), open path infrared gas analyzer, and methane flux sensor (LI-COR, Lincoln, NE, LI7700). There was a different radiation sensor at US-Los (Kipp & Zonen 197 North America, Sterling, USA, Kipp-Zonen CNR4) than US-ALQ (Apogee Instruments Inc., Logan, UT, SN-500). There was also a different air temperature and relative humidity sensor at US-Los (Campbell Scientific, Inc., Logan, UT, CS215) than at US-ALQ (Campbell Scientific, Inc., Logan, UT, Vaisala HMP45C 190 platinumresistance thermometer). A quantum photosynthetically active radiation (PAR) sensor was installed only at US-Los (LI-COR, Lincoln, NE, LI-190). More information on flux tower height, footprint, and instrumentation for both sites can be found in Turner et al. (2019).

Typical in eddy covariance studies, there were portions of data missing from the continuous record that required gap-filling. There were more missing half-hourly FCH<sub>4</sub> data at US-ALQ than US-Los (49% vs. 34%). Fewer data were missing during the typical CO<sub>2</sub> uptake period from April to October (US-ALQ 44%, US-Los 23%) than during the rest of the year for both sites (US-ALQ 57%, US-Los 48%). There were more data gaps in 2019 (US-ALQ 63%, US-Los 45%) than 2020 (US-ALQ 36%, US-Los 22%). FCH<sub>4</sub> was gap-filled with the machine learning random forest algorithm using the "randomForest" R package (Liaw and Wiener, 2001) and gap-filling script (Kim, 2020; Kim et al., 2020). This approach was selected because it outperforms marginal distribution sampling, artificial neural networks, and support vector machine for gap-filling of eddy covariance FCH<sub>4</sub> data.

Net ecosystem exchange of CO<sub>2</sub> (NEE) was gap-filled and partitioned into GPP and ecosystem respiration (R<sub>eco</sub>) using the Desai-Cook flux partitioning model (Cook et al., 2004; Desai et al., 2007), which utilizes a non-linear regression of daytime R<sub>eco</sub> to PAR and is comparable to many other regressions based on moving window flux partitioning algorithms. R<sub>eco</sub> in the model is calculated from a non-linear regression of nighttime NEE to T<sub>air</sub>. Once again, there were more data gaps in 2019 (US-ALQ 37%, US-Los 49%) than in 2020 (US-ALQ 29%, US-Los 33%), for NEE. Fewer data were missing during the CO<sub>2</sub> uptake period from April to October (US-ALQ 29%, US-Los 34%) than during the rest of the year (US-ALQ 38%, US-Los 51%). No missing data remained after gap-filling NEE for US-ALQ during either year. No

missing data remained after gap-filling NEE for US-Los in 2019, but a small amount of missing data remained after gap-filling in 2020 (US-Los 94%).

## 2.3 Meteorological and hydrological data

Daily total precipitation data were from Lakeland Field Station at Lakeland Airport in Woodruff, WI (45.927222, -89.730836), located 26.1 km away from US-Los and 15.4 km from US-ALQ (NOAA, 2021). Rhinelander, WI, a city located 56 km from the study sites, received nearly 110 cm of total precipitation in 2019, making it the wettest year on record from 1908 to 2020 (Rhinelander Weather Recs., 2021). The average annual precipitation for Rhinelander during that time was  $80 \pm 15$  cm (standard deviation). Meanwhile, the city received only 90 cm of precipitation in 2020. Stream temperature and temperature data for US-ALQ were from the National Water Information System (USGS, 2020a; https://waterdata.usgs.gov/nwis). US-ALQ lacked water table depth data during the time of study. WTD was used in place of stream discharge to understand the impact of hydrology on FCH<sub>4</sub> at US-Los. The water level sensor used to measure WTD was a Campbell Scientific CS451.

#### 2.4 GPP-FCH<sub>4</sub> Lag analysis

Lag analysis involved multiple steps. First, we utilized a built-in function in MATLAB that performed a circular shift of the data ("xcorr") to estimate the direction of the strongest lag correlation based on cross correlation of the two variables of interest. Lag analysis was then performed in the direction of the strongest lag correlation as in Rinne et al. (2018). The driver variable was lagged with respect to the response variable one step at a time. Rows with missing driver variable data were removed. Correlation and significance between variables were measured at each step. Variables analyzed for lags included GPP & FCH<sub>4</sub>, GPP & T<sub>air</sub>-FCH<sub>4</sub>, WTD & FCH<sub>4</sub>, WTD & T<sub>air</sub>-FCH<sub>4</sub>, and WTD & GPP. The lag was not measured past 200 days. MATLAB scripts for creating the lag analysis plots and all other plots used in this study are available online (turner-j, 2020). All lag analysis was performed with daily average data.

#### 2.5 Temperature normalization of FCH<sub>4</sub>

Covariates must be taken into careful consideration when analyzing data for FCH<sub>4</sub> drivers that are interrelated. One such covariate that has a noticeable impact on FCH<sub>4</sub> is temperature, but whether it is the soil, air, or water temperature which dominantly influences FCH<sub>4</sub> varies among

sites (Rey-Sanchez et al., 2018; Chu et al., 2014). Removing the influence of temperature on FCH<sub>4</sub> before performing lag analysis with GPP should be done to eliminate any trends solely due to temperature as a driving force (Chen et al., 2020). We removed the influence of  $T_{air}$  on FCH<sub>4</sub> by fitting observed FCH<sub>4</sub> to an exponential model with  $T_{air}$  as a predictor. Non-linear regression was used to find the coefficients  $\beta_1$  and  $\beta_2$  in equation (1) below. Observed FCH<sub>4</sub> from both years was then divided by predicted FCH<sub>4</sub> based on observed  $T_{air}$  as in equation (2) below.  $T_{air}$  dependence of FCH<sub>4</sub> was therefore modeled and removed using the following equations:

(1) 
$$F = \beta_1 \times \exp(\beta_2 \cdot T)$$
  
(2)  $T_{air}FCH_4 = \frac{daily average FCH_4}{E}$ 

#### 3 Results

3.1 Meteorology and stream discharge

The two study sites, US-ALQ and US-Los, had equal means of daily average incoming radiation (p = 0.054) and linearly correlated but unequal means of air temperature (r = 1,  $p \ll 0.01$ ) and vapor pressure deficit (r = 0.97,  $p \ll 0.01$ ). Resulting differences in the variables we compare and analyze in this study at each site could therefore be due to the impacts of mesoscale or microscale meteorology; differences in magnitude of stream discharge, air temperature, or vapor pressure deficit; or abiotic or biotic site characteristics other than those previously mentioned.

The water table at US-Los was closer to the surface in 2019 than 2020 (-0.26 vs. -0.29 m below surface). Additionally, stream discharge at US-ALQ was linearly related to WTD at US-Los (r = 0.45, p = << 0.01) and was slightly higher in 2019 (0.15 m<sup>3</sup>s<sup>-1</sup>) than 2020 (0.16 m<sup>3</sup>s<sup>-1</sup>). ). There was a low covariance (cov = 0.0045) and a significant, positive two-day lagged effect of precipitation at Lakeland Airport on WTD at US-Los (r = 0.11). Precipitation covaried with stream discharge at US-ALQ (cov = 0.01). Precipitation also covaried with T<sub>stream</sub> at US-ALQ (cov = 0.56) and T<sub>air</sub> at US-Los (cov = 0.81) and US-ALQ (cov = 0.91).

3.2 CH<sub>4</sub> and CO<sub>2</sub> fluxes in wet (2019) and dry (2020) years

FCH<sub>4</sub> gap-filling performance was higher with US-ALQ than US-Los ( $R^2 = 0.85$  vs. 0.70). No FCH<sub>4</sub> data were missing at either site after gap-filling (Fig 1). US-ALQ emitted more than double the daily average FCH<sub>4</sub> as US-Los during the entire study period (approx. 19.6 versus 6.7 µmol m<sup>-2</sup> s<sup>-1</sup>). Daily average FCH<sub>4</sub> reached a maximum of 47.7 µmol CH<sub>4</sub> m<sup>-2</sup> s<sup>-1</sup> at US-Los and 117.9 µmol CH<sub>4</sub> m<sup>-2</sup> s<sup>-1</sup> at US-ALQ. More carbon as CH<sub>4</sub> (C-CH<sub>4</sub>) was emitted and less carbon as CO<sub>2</sub> (C-CO<sub>2</sub>) was taken up annually at US-ALQ than US-Los in both years (Table 1). Both sites exhibited lower daily mean FCH<sub>4</sub> in the historically wet year of 2019 (5.5 µmol CH<sub>4</sub> m<sup>-2</sup>s<sup>-1</sup> US-Los; 13.5 µmol CH<sub>4</sub> m<sup>-2</sup>s<sup>-1</sup> US-ALQ) than in 2020 (7.8 µmol CH<sub>4</sub> m<sup>-2</sup>s<sup>-1</sup> US-Los; 25.8 µmol CH<sub>4</sub> m<sup>-2</sup>s<sup>-1</sup> US-ALQ). Cumulative annual C-CH<sub>4</sub> emission at both sites was only a fraction of C-CO<sub>2</sub> uptake. US-ALQ had the highest cumulative annual FCH<sub>4</sub> in comparison to FCO<sub>2</sub>, at 10.31% in 2020. US-Los had the lowest cumulative annual FCH<sub>4</sub> in comparison to FCO<sub>2</sub>, at approximately 1% in 2019.



**Fig 1.** (A) Bar plot of precipitation at Lakeland Airport and timeseries of water table depth (WTD) at US-Los, (B) stream discharge (Q) at US-ALQ, (C) air temperature in degrees C at both sites and stream water temperature at US-ALQ, (D) GPP at both sites in  $\mu$ mol CO<sub>2</sub> m<sup>-2</sup>s<sup>-1</sup>, and (E) FCH<sub>4</sub> at both sites in  $\eta$ mol CH<sub>4</sub> m<sup>-2</sup>s<sup>-1</sup> and historic average FCH<sub>4</sub> at US-Los from 2014 to 2018.

**Table 1.** Cumulative annual FCH<sub>4</sub> and FCO<sub>2</sub> as C from US-ALQ and US-Los in 2019 and 2020.

	FCH <sub>4</sub> (g C-CH <sub>4</sub> m <sup>-2</sup> yr <sup>-1</sup> )		$FCO_2 (g C-CO_2 m^{-2} yr^{-1})$	
	2019	2020	2019	2020
US-ALQ	$5.08 \pm 0.10$	$9.79 \pm 0.21$	$-78.36 \pm 2.59$	$-94.92 \pm 2.52$
US-Los	$2.08 \pm 0.04$	$2.98 \pm 0.06$	$-202.61 \pm 5.79$	$-148.73 \pm 4.69$

*Note:* Values are shown with standard error of the mean. Negative values indicate a C sink from the atmosphere. Positive values indicate a C source to the atmosphere.

## 3.3 GPP-FCH<sub>4</sub> relationship

FCH<sub>4</sub> preceded GPP by approximately 20 days at US-Los in 2019 (Fig 2A). Removing the influence of air temperature ("normalized" in Figs 2,3) resulted in a stronger lagged correlation where FCH<sub>4</sub> followed GPP starting after 40 days in 2019, and the strength of the correlation continued to grow up until at least 100 days (Fig 2B). The lag relationships between FCH<sub>4</sub> and GPP, as well as  $T_{air}$ -FCH<sub>4</sub> and GPP, were both weakly correlated in 2019 (r < 0.4). The lagged influence of GPP on FCH<sub>4</sub> at US-Los peaked at 40 days in 2020 (Fig 2C). The lagged influence of GPP on  $T_{air}$ -FCH<sub>4</sub> at US-Los was shorter than that of FCH<sub>4</sub> in 2020, with a broad peak lasting 40-60 days (r = 0.65).



**Fig 2.** Lagged influence of GPP on FCH<sub>4</sub> and temperature-normalized FCH<sub>4</sub> at US-Los. (A) FCH<sub>4</sub> preceding GPP in 2019. (B) FCH<sub>4</sub> following GPP in 2019. (C) FCH<sub>4</sub> following GPP in 2020. Positive values indicate FCH<sub>4</sub> follows GPP. Only significant (p < 0.05), positive lags (r > 0) shown.

GPP preceded FCH<sub>4</sub> by approximately 20 days at US-ALQ in 2019 (Fig 3A). Removing the influence of air temperature on FCH<sub>4</sub> revealed that the strongest correlation between GPP and  $T_{air}$ -FCH<sub>4</sub> occurred around 60 days and plateaued until at least 100 days, meaning that GPP was not correlated with FCH<sub>4</sub> for at least 2 months (r = 0.52). The lag relationship was slightly shorter in 2020, with GPP leading FCH<sub>4</sub> by roughly 20 days and GPP leading  $T_{air}$ -FCH<sub>4</sub> by approximately 35-50 days (r = 0.78, Fig 3B). The lagged influence of GPP on  $T_{air}$ -FCH<sub>4</sub> was shorter, stronger, and more closely related to that of GPP and FCH<sub>4</sub> in 2020 than 2019 for both sites.



**Fig 3.** Lagged influence of GPP on FCH<sub>4</sub> and temperature-normalized FCH<sub>4</sub> at US-ALQ in 2019 and 2020. Only significant (p < 0.05), positive lags (r > 0) shown.

3.4 Influences of hydrology

FCH<sub>4</sub> at both sites was correlated with  $T_{air}$ , but the relationship was stronger at US-ALQ ( $R^2 = 0.53$ ) than US-Los ( $R^2 = 0.25$ ) (Figs 4A & 5A). The relationship between WTD and  $T_{air}$ -FCH<sub>4</sub> at US-Los was not linear in either year (Fig 4B & C). There was a significant inverse relationship between discharge and  $T_{air}$ -FCH<sub>4</sub> at US-ALQ during 2019 ( $R^2 = 0.73$ ) (Fig 5B). However, there was no significant relationship between discharge and  $T_{air}$ -FCH<sub>4</sub> at US-ALQ in 2020 (Fig 5C).



**Fig 4.** (A) Air temperature sensitivity of FCH<sub>4</sub> in both years combined, and (B) WTD versus temperature-normalized FCH<sub>4</sub> using daily average data at US-Los during 2019 and (C) 2020.  $R^2$  displayed on the first subplot is the coefficient of determination between the best fit of the relevant model (second order polynomial) and observations. Best fit lines show significant relationships (p < 0.05). T<sub>air</sub>- FCH<sub>4</sub> was bin averaged according to discharge rounded to the nearest 0.01 m.



**Fig 5.** (A) Air temperature sensitivity of FCH<sub>4</sub> in both years and (B) Discharge versus  $T_{air}$ - FCH<sub>4</sub> using daily average data at US-ALQ during 2019 and (C) 2020. R<sup>2</sup> displayed on each plot is the coefficient of determination between the best fit of the relevant model (first or second order polynomial) and observations. Best fit lines show significant relationships (p < 0.05).  $T_{air}$ - FCH<sub>4</sub> was bin averaged according to discharge rounded to the nearest 0.01 m<sup>3</sup>s<sup>-1</sup>.

There was a significant lagged influence of WTD on FCH<sub>4</sub> at US-Los in 2019 (Fig 6A). Removing the influence of  $T_{air}$  did not greatly change the relationship (Fig 6B). The lagged effect of WTD on  $T_{air}$ -FCH<sub>4</sub> lasted 5-52 days with a peak at 24 days and a correlation coefficient of 0.27. No significant, positive correlation was detected between WTD and FCH<sub>4</sub> or  $T_{air}$ -FCH<sub>4</sub> at US-Los during 2020 (data not shown).



**Fig 6.** Lagged effects of water table depth (WTD) on (A) FCH<sub>4</sub> and (B)  $T_{air}$ -FCH<sub>4</sub> at US-Los in 2019. Filled blue circles represent significant (p < 0.05), positive (r > 0) lag correlations. Empty circles are not significant and/or do not represent positive correlations.

WTD variability at US-Los was lower in 2020 (var = 0.002) than 2019 (var = 0.016). Discharge variance was consistently low at US-ALQ, remaining at 0.002 in both years.

Discharge was also low but typical for the stream, ranging from 0.086-0.32 m<sup>3</sup>s<sup>-1</sup> in 2019 and from 0.10-0.46 m<sup>3</sup>s<sup>-1</sup> in 2020 at US-ALQ. There was also a significant, positive (p< 0.05, r>0) lagged effect of WTD on GPP at US-Los with a very long duration of 43-140 days with a peak at 92 days (r = 0.81) in 2019 (Fig 7A). The lag effect was similar in 2020 but did not begin until around 60 days and had a lower peak correlation (r = 0.42, Fig 7B).



**Fig 7.** Lagged effects of water table depth (WTD) on GPP in (A) 2019 and (B) 2020 at US-Los. Filled blue circles represent significant (p < 0.05), positive (r > 0) lag correlations. Empty circles are not significant and/or do not represent positive correlations.

Higher  $T_{stream}$  was significantly positively correlated (r = 0.15, p = 0.003) with higher  $T_{air}$ -FCH<sub>4</sub> at US-ALQ from April to October but negatively correlated with  $T_{air}$ -FCH<sub>4</sub> during the rest of the year (r = -0.28, p = 0) during both years combined (Fig 8 A & B). However, the linear model demonstrated a poor fit to the data in both cases (R<sup>2</sup> < 0.1). Stream temperature did not surpass 5 °C during months outside April to October. Average yearly water temperature in the creek was slightly lower in 2019 than 2020 (9.3 vs. 9.6 °C).



**Fig 8.** The influence of  $T_{stream}$  on  $T_{air}$ -FCH<sub>4</sub> at US-ALQ during (A) April to October and (B) all other months for both years combined. Solid lines represent the first-order linear regression.  $R^2$  is the coefficient of determination between the linear regression and observations.

#### Discussion

4.1 CH<sub>4</sub> limitations in a historically wet year

Daily average FCH<sub>4</sub> for both sites was on the lower end of what is expected for wetlands in general (Nicolini et al., 2013), but cumulative annual FCH<sub>4</sub> aligned well within the wetland type of fens (Knox et al., 2019). Daily average FCH<sub>4</sub> and cumulative annual FCH<sub>4</sub> for both sites were lower in the historically wet year than in the following dry year (Table 1, Figure 1E). This is a noteworthy finding given that an earlier study in the same region, but from a landscape-level tall tower, concluded a dry year with a longer growing season and warmer conditions cut FCH<sub>4</sub> by 28% (Desai et al., 2015). These decreases in FCH<sub>4</sub> in unusually wet and dry years suggest environmental extremes could reduce cumulative annual FCH<sub>4</sub> at these sites. The results of this study also strengthen the idea of a critical inundation level past which wetland CH<sub>4</sub> emissions begin to decline due to a number of possible reasons (e.g., lower light attenuation, diluted organic substrate, etc.) (Calabrese et al., 2021). However, more hydrological data from both sites is needed.

## 4.2 GPP-FCH<sub>4</sub> lagged relationship

At both sites, the lag between GPP and FCH<sub>4</sub> was weaker and took longer in 2019 than in 2020. The GPP-FCH<sub>4</sub> lag relationship observed at both sites during the study is supported by Delwiche et al. (2021), which found a lag relationship between FCH<sub>4</sub> and GPP in 83% of global freshwater wetlands, and the 20.7-day lag observed in Knox et al. (2021). T<sub>air</sub> normalization of FCH<sub>4</sub> was critical for observation of the severe shifts in the lag effect in this study during a historically wet year, as normalization shifted the GPP-FCH<sub>4</sub> relationship at US-ALQ backwards by roughly twenty days in 2019. The lag correlation between GPP and subsequent T<sub>air</sub>-FCH<sub>4</sub> observed at both sites in 2019 superseded the amount of time needed for photosynthesis, soil C fixation, and root growth, indicating the influence of another factor not considered in this study or a process that needs to be further explored.

The extremely long lagged influences of GPP on  $T_{air}$ -FCH<sub>4</sub> at both sites, and WTD on GPP at US-Los in comparison with the much shorter lag between WTD and FCH<sub>4</sub>, indicate that root respiration was not a strong driver of FCH<sub>4</sub> in the beginning of 2019. One possible explanation is substrate limitation for methanogenesis caused by a lack of recently fixed labile C or older, more recalcitrant soil organic carbon (Oikawa et al., 2017).

In a year characterized by historical levels of precipitation, WTD fluctuated from approximately 0.2 m to -0.6 m from May to August of 2019 at US-Los. This brings into question the potential of plant stress to limit GPP, as high water level fluctuations will increase plant biomass allocation to roots rather than shoots and can have a negative impact on propagation (Wei et al., 2019) and photosynthetic potential (Ballantyne et al., 2014). Additionally, in flood years following droughts, maximum growing season GPP could decline due to plant stress or change in vegetation composition (Olefeldt et al., 2017), and the expected consequential increase in FCH4 would not occur. Plant stress response and the resulting impact on gas flux is an area worthy of further research especially as extreme precipitation becomes more common.

4.3 WTD

Pugh et al. (2018) investigated WTD and monthly average FCH<sub>4</sub> at US-Los and found no correlation when accounting for T<sub>air</sub>, but our analysis of daily average fluxes in a historically wet year revealed a lag effect of the two variables that lasted 5-52 days but reached peak correlation at 24 days. Peak lag correlation of WTD and FCH<sub>4</sub> in 2019 aligned well with other sites from the FLUXNET-CH4 database, which averaged approximately 18.3 days (Knox et al., 2021). The response of GPP did not follow until nearly 40 days later at US-Los in 2019, a year that was characterized by historic precipitation.

High water table level will reduce seedling establishment, growth, and survival in wetlands if it occurs during seedling establishment or for a prolonged period of time (Zacks et al., 2019). Although some plants are more resilient to flooded conditions, permanently flooded conditions cause oxygen deprivation and higher CO<sub>2</sub> storage in plant tissues and at the cellular level (Pedersen et al., 2017). A shallower water table (i.e., closer to the surface) should increase FCH<sub>4</sub> by increasing GPP of hydric vegetation (Gomez-Casanovas et al., 2020; Musarika et al., 2017). Other studies disagree, demonstrating that a deeper water table (i.e., farther below the surface) will increase GPP in the absence of moisture stress by improving the availability of O<sub>2</sub> for photosynthesis in roots (Ballantyne et al., 2014).

Analysis of twenty-three sites from the FLUXNET-FCH4 database has shown that  $T_{air}$  controls FCH<sub>4</sub> at sites with lower WTD variability, but WTD controls FCH<sub>4</sub> at sites with lower  $T_{air}$  variability (Knox et al., 2019; Delwiche et al., 2021). The lack of correlation between lagged WTD and FCH<sub>4</sub> at US-Los in 2020, a year with more WTD variability, supports this. However, the positive relationship between WTD variability and FCH<sub>4</sub> appears to be species-specific (Radu and Duval, 2018).

4.4 Study limitations

Among the variables considered in this study, there are expected covariate relationships between precipitation,  $T_{air}$ ,  $T_{stream}$ , WTD, and stream discharge, and seasonal cycles for each. WTD and temperature may alter the GPP-FCH<sub>4</sub> relationship because their covariance can appear like a cause and effect (direct relationship) when it is instead evidence of an indirect relationship. For example, the negative relationship between  $T_{stream}$  and  $T_{air}$ -FCH<sub>4</sub> outside of the months April-October (Fig 8B) demonstrated how shifting seasonal patterns may mask the relationship of GPP and FCH<sub>4</sub>.

The interactions of precipitation and  $T_{air}$  are evidenced in Dinsmore et al. (2013), where essentially all interannual variability in the export of dissolved organic carbon from a peatland catchment was explained by interactions between the two variables. Additionally, interannual variability of total aquatic carbon (POC, DOC, DIC) concentration in a stream draining a peatland was strongly connected to GPP, but the main source of evaded CO<sub>2</sub> (unclear whether from stream or wetland) was suspected to be deep within the soil profile and disconnected from surface processes to some extent. This supports the idea that terrestrially-derived CO<sub>2</sub> in groundwater is a dominant source (other than soil) of total dissolved gas flux from riverine and wetland ecosystems (Olde, 2017). Other studies have aligned with a deep soil source (below 20 cm) of CH<sub>4</sub> as well (Peng et al., 2017). Flow regime and soil water content also has a clear impact on instream and riparian GPP within a forested biome (Dodd, 2018). Stream discharge and WTD should therefore be carefully considered in comparative FCH<sub>4</sub> driver analysis due to their interactions.

Although data were gap-filled, the fraction of missing or low-quality data that was removed was typical of eddy covariance flux data. Gap filling did not appreciably change conclusions. Where results did change as a result of gap-filling, it was the product of unequal sample sizes across years.

4.5 Future work

Further research at US-ALQ and US-Los could help pinpoint the magnitude and extent of FCH<sub>4</sub> from wetlands and wetland streams and quantify the variability in wetland FCH<sub>4</sub> in closely located sites due to random effects, such as differing microbial communities and their resulting rates of methanogenesis. Stream discharge and  $T_{stream}$  are tied to wetland FCH<sub>4</sub> but have strong spatial variability. Taking more frequent measurements of these variables and sampling different locations within the wetland and within the water column or peat profile may reveal relationships that were previously masked by spatial or temporal variability.

#### 5 Conclusions

Here, we presented FCH<sub>4</sub> and corresponding hydrological measurements from two wetland sites to determine (1) the importance of plant C fixation measured by the lagged effect of GPP on FCH<sub>4</sub> and (2) how factors relating to wetland hydrology (i.e., stream discharge,

T<sub>stream</sub>, and WTD) mediate the GPP-FCH<sub>4</sub> relationship. This study showed that two closely located wetlands can produce vastly different FCH<sub>4</sub> and demonstrate different seasonal cycles of FCH<sub>4</sub> because of different plant and microbial communities and responses, especially during a year with extreme precipitation. During a year with historically high precipitation, there was lower cumulative annual and daily average FCH<sub>4</sub> from the wetlands compared to the following drier year. Both wetlands displayed a longer lagged effect of GPP on FCH<sub>4</sub> during the wet year. US-ALQ demonstrated a decrease in T<sub>air</sub>-FCH<sub>4</sub> with increasing stream discharge in 2019 and not in 2020, but US-Los exhibited no significant linear trend between WTD and T<sub>air</sub>-FCH<sub>4</sub> in either year unless a lag was introduced. Lag analysis showed that FCH<sub>4</sub> response to WTD preceded GPP response to WTD at US-Los. A potential explanation is microbial respiration was more reliant on preexisting soil organic matter as a C source earlier in the season but was sourced by recently fixed plant C later in the season.

To answer to the second question of our study, we considered indicators of wetland hydrology and analyzed their relationship with FCH<sub>4</sub>. As previously mentioned, there was a shorter lagged response of FCH<sub>4</sub> to WTD than GPP to WTD at US-Los. However, there was no relationship between FCH<sub>4</sub> and WTD in 2020. The analysis of stream discharge or WTD alone can potentially mask the influence of groundwater flow or precipitation on T<sub>water</sub> and resulting daily average FCH<sub>4</sub>. It was important to consider the influence of T<sub>stream</sub> during the on-season and off-season separately, and to remove the influence of T<sub>air</sub> on FCH<sub>4</sub> when looking at the impact of stream discharge. Questions remain on whether larger fluctuations in WTD caused or indicated conditions that could have caused plant or microbial stress and lowered FCH<sub>4</sub> during 2019 at US-Los in comparison to US-ALQ, which emitted more CH<sub>4</sub> and displayed a stronger seasonal cycle. Additional work of linking lags between productivity and FCH<sub>4</sub> along with accounting for temperature and discharge effects will help clarify and constrain the role of wetland biogeochemistry in a changing (e.g., wetter or drier) climate.

#### Acknowledgements and Data

We thank our reviewers and Dr. Gavin McNicol for their valuable comments on our manuscript, and Dr. Jaclyn Matthes and Dr. Janne Rinne for providing more information on their related studies. This material is based upon work supported by the National Science Foundation

Graduate Research Fellowship Program under Grant No. DGE-1747503. Any opinions, findings, and conclusions or recommendations expressed in this material are those of the author(s) and do not necessarily reflect the views of the National Science Foundation. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government. We also acknowledge the North Temperate Lakes LTER program (NSF DEB654 1440297, NTL LTER), United States Geological Survey (Water, Energy, and Biogeochemical Budgets Program; LandCarbon Program; Water Mission Area), and the Department of Energy Ameriflux Network Management Program. Datasets for this research are available in these intext data citation references: (Olson, B., 2020; Desai, A. 2020). The authors do not declare any conflicts of interest.

### References

Anderson, M., and Lowry, C. (2007). Transient Functioning of a Groundwater Wetland Complex, 662 Allequash Basin, Wisconsin. PDF File. www.wri.wisc.edu/wpcontent/uploads/FinalWR05R007.pdf

Ballantyne, D. M., Hribljan, J. A., Pypker, T. G., & Chimner, R. A. (2014). Long-term water table manipulations alter peatland gaseous carbon fluxes in Northern Michigan. *Wetlands ecology and management*, 22(1), 35-47.

Bridgham, S. D., Cadillo- Quiroz, H., Keller, J. K., & Zhuang, Q. (2013). Methane emissions from wetlands: biogeochemical, microbial, and modeling perspectives from local to global scales. *Global change biology*, *19*(5), 1325-1346. DOI: 10.1111/gcb.12131

Calabrese, S., Garcia, A., Wilmoth, J. L., Zhang, X., & Porporato, A. (2021). Critical inundation level for methane emissions from wetlands. *Environmental Research Letters*. https://iopscience.iop.org/article/10.1088/1748-9326/abedea/meta Centeno, C. A. R., Alberto, M. C. R., Wassmann, R., & Sander, B. O. (2017). Assessing diel variation of CH4 flux from rice paddies through temperature patterns. *Atmospheric environment*, *167*, 23-39. https://doi.org/10.1016/j.atmosenv.2017.08.007

Chen, W., Wang, B., Zhang, F., Li, Z., Wang, J., Yu, G., et al. (2020). Hysteretic relationship between plant productivity and methane uptake in an alpine meadow. *Agricultural and Forest Meteorology*, 288, 107982.

https://www.sciencedirect.com/science/article/pii/S0168192320300848

Chu, H., Chen, J., Gottgens, J. F., Ouyang, Z., John, R., Czajkowski, K., & Becker, R. (2014). Net ecosystem methane and carbon dioxide exchanges in a Lake Erie coastal marsh and a nearby cropland. *Journal of Geophysical Research: Biogeosciences*, *119*(5), 722-740. https://doi.org/10.1002/2013JG002520

Cook, B. D., Davis, K. J., Wang, W., Desai, A., Berger, B. W., Teclaw, R. M., ... & Heilman, W. (2004). Carbon exchange and venting anomalies in an upland deciduous forest in northern Wisconsin, USA. *Agricultural and Forest Meteorology*, *126*(3-4), 271-295. https://doi.org/10.1016/j.agrformet.2004.06.008

Dannenberg, S., & Conrad, R. (1999). Effect of rice plants on methane production and rhizospheric metabolism in paddy soil. *Biogeochemistry*, *45*(1), 53-71. https://link.springer.com/article/10.1007/BF00992873

Davidson, S. J., Sloan, V. L., Phoenix, G. K., Wagner, R., Fisher, J. P., Oechel, W. C., & Zona, D. (2016). Vegetation type dominates the spatial variability in CH 4 emissions across multiple arctic tundra landscapes. *Ecosystems*, *19*(6), 1116-1132. https://link.springer.com/article/10.1007/s10021-016-9991-0

Delwiche, K. B., Knox, S. H., Malhotra, A., Fluet-Chouinard, E., McNicol, G., Feron, S., ... & Jackson, R. B. (2021). FLUXNET-CH4: A global, multi-ecosystem dataset and analysis of methane seasonality from freshwater wetlands. *Earth System Science Data Discussions*, 1-111. https://doi.org/10.5194/essd-2020-307

Desai, A. R., Xu, K., Tian, H., Weishampel, P., Thom, J., Baumann, D., ... & Kolka, R. (2015). Landscape-level terrestrial methane flux observed from a very tall tower. *Agricultural and Forest Meteorology*, 201, 61-75. https://doi.org/10.1016/j.agrformet.2014.10.017

Desai, A.R. (2001-). AmeriFlux US-Los Lost Creek, Dataset. https://doi.org/10.17190/AMF/1246071

Desai, A., Noormets, A., Bolstad, P., Chen, J., Cook, B., Curtis, P., Davis, K., Euskirchen, E., Gough, C., Martin, J., Ricciuto, D., Schmid, H., Su, H-B., Tang, J., Vogel, C., and Wang, W. (2007). Influence of vegetation type, stand age and climate on carbon dioxide fluxes across the Upper Midwest, USA: implications for regional scaling of carbon flux.

https://www.researchgate.net/publication/228990606\_Influence\_of\_vegetation\_type\_stand\_age\_ and\_climate\_on\_carbon\_dioxide\_fluxes\_across\_the\_Upper\_Midwest\_USA\_implications\_for\_re gional\_scaling\_of\_carbon\_flux

Ding, W., Cai, Z., & Tsuruta, H. (2005). Plant species effects on methane emissions from freshwater marshes. *Atmospheric Environment*, *39*(18), 3199-3207. https://doi.org/10.1016/j.atmosenv.2005.02.022

Dinsmore, K.J., Billett, M.F. and Dyson, K.E. (2013). Temperature and precipitation drive temporal variability in aquatic carbon and GHG concentrations and fluxes in a peatland catchment. *Glob Change Biol*, 19: 2133-2148. https://doi.org/10.1111/gcb.12209

Dodd, A. (2018). Flow Regime Influences on Stream and Riparian Soil Carbon Dynamics in the Ozark Highlands and Boston Mountains of Arkansas. https://scholarworks.uark.edu/cgi/viewcontent.cgi?article=4460&context=etd

Dorodnikov, M., Knorr, K. H., Kuzyakov, Y., & Wilmking, M. (2011). Plant-mediated CH 4 transport and contribution of photosynthates to methanogenesis at a boreal mire: a 14 C pulse-labeling study. *Biogeosciences*, 8(8), 2365-2375. https://doi.org/10.5194/bg-8-2365-2011

Gomez- Casanovas, N., DeLucia, N. J., DeLucia, E. H., Blanc- Betes, E., Boughton, E. H., Sparks, J., & Bernacchi, C. J. (2020). Seasonal controls of CO2 and CH4 dynamics in a temporarily flooded subtropical wetland. *Journal of Geophysical Research: Biogeosciences*, *125*(3), e2019JG005257. https://doi.org/10.1029/2019JG005257

Hatala, J. A., Detto, M., & Baldocchi, D. D. (2012). Gross ecosystem photosynthesis causes a diurnal pattern in methane emission from rice. *Geophysical Research Letters*, *39*(6). https://doi.org/10.1029/2012GL051303

Henneberg, A., Sorrell, B. K., & Brix, H. (2012). Internal methane transport through J uncus effusus: experimental manipulation of morphological barriers to test above- and below- ground diffusion limitation. *New Phytologist*, *196*(3), 799-806. https://doi.org/10.1111/j.1469-8137.2012.04303.x

Kim, Y., Johnson, M. S., Knox, S. H., Black, T. A., Dalmagro, H. J., Kang, M., ... & Baldocchi,
D. (2020). Gap- filling approaches for eddy covariance methane fluxes: A comparison of three machine learning algorithms and a traditional method with principal component analysis. *Global Change Biology*, 26(3), 1499-1518. https://doi.org/10.1111/gcb.14845

Kim, Y. (2020). Repository EC\_FCH4\_gapfilling, script rf\_run\_for\_public. https://github.com/yeonukkim/EC\_FCH4\_gapfilling/blob/master/rf\_run\_for\_public.R

Knox, S.H., Jackson, R.B., Poulter, B., McNicol, G. Fluet-Chouinard, E., Zhang, Z., Hugelius, G., et al. (2019). "FLUXNET-CH4 Synthesis Activity: Objectives, Observations, and Future Directions." *Bulletin of the American Meteorological Society* 100 (12): 2607–32. https://doi.org/10.1175/BAMS-D-18-0268.1

Knox, S. H., Bansal, S., McNicol, G., Schafer, K., Sturtevant, C., Ueyama, M., ... & Jackson, R.B. (2021). Identifying dominant environmental predictors of freshwater wetland methane fluxes across diurnal to seasonal time scales. *Global change biology*.

Kuzyakov, Y., & Gavrichkova, O. (2010). Time lag between photosynthesis and carbon dioxide efflux from soil: a review of mechanisms and controls. *Global Change Biology*, *16*(12), 3386-3406. https://doi.org/10.1111/j.1365-2486.2010.02179.x

Liaw, A., & Wiener, M. (2002). Classification and regression by randomForest. *R News*, **2**(3), 18–22.

https://www.researchgate.net/publication/228451484\_Classification\_and\_Regression\_by\_Rando mForest

Mitra, B., Minick, K., Miao, G., Domec, J. C., Prajapati, P., McNulty, S. G., ... & Noormets, A. (2020). Spectral evidence for substrate availability rather than environmental control of methane emissions from a coastal forested wetland. *Agricultural and Forest Meteorology*, *291*, 108062. https://doi.org/10.1016/j.agrformet.2020.108062

Musarika, S., Atherton, C. E., Gomersall, T., Wells, M. J., Kaduk, J., Cumming, A. M. J., ... & Zona, D. (2017). Effect of water table management and elevated CO2 on radish productivity and

Olson, B. (2015-). AmeriFlux US-ALQ Allequash Creek Site, Dataset. https://doi.org/10.17190/AMF/1480323

Peng, H., Guo, Q., Hong, B., Ding, H., Xu, C., & Yao, H. (2017). Seasonal and diurnal methane emissions from a wetland meadow on the Eastern Qinghai-Tibetan Plateau: effects of soil temperature, water table level and gross primary productivity (GPP). In *EGU General Assembly Conference Abstracts* (p. 8004). https://ui.adsabs.harvard.edu/abs/2017EGUGA..19.8004P/abstract

Pint, C. D., Hunt, R. J., & Anderson, M. P. (2003). Flowpath delineation and ground water age, Allequash Basin, Wisconsin. *Groundwater*, *41*(7), 895-902. https://doi.org/10.1111/j.1745-6584.2003.tb02432.x

Pugh, C. A., Reed, D. E., Desai, A. R., & Sulman, B. N. (2018). Wetland flux controls: how does interacting water table levels and temperature influence carbon dioxide and methane fluxes in northern Wisconsin? *Biogeochemistry*, *137*(1), 15-25. https://link.springer.com/article/10.1007/s10533-017-0414-x

Radu, D. D., & Duval, T. P. (2018). Impact of rainfall regime on methane flux from a cool temperate fen depends on vegetation cover. *Ecological Engineering*, *114*, 76-87. https://doi.org/10.1016/j.ecoleng.2017.06.047

Rey-Sanchez, A. C., Morin, T. H., Stefanik, K. C., Wrighton, K., & Bohrer, G. (2018). Determining total emissions and environmental drivers of methane flux in a Lake Erie estuarine marsh. *Ecological Engineering*, *114*, 7-15. https://doi.org/10.1016/j.ecoleng.2017.06.042

Rhinelander Weather Records. (2021). Retrieved from

https://www.weather.gov/media/grb/climate/books/Rhinelander%20Record%20Books2.pdf

Rietl, A. J., Nyman, J. A., Lindau, C. W., & Jackson, C. R. (2017). Wetland methane emissions altered by vegetation disturbance: An interaction between stem clipping and nutrient enrichment. *Aquatic Botany*, *136*, 205-211. https://doi.org/10.1016/j.aquabot.2016.10.008

Rinne, J., Tuittila, E.- S., Peltola, O., Li, X., Raivonen, M., Alekseychik, P., et al. (2018). Temporal variation of ecosystem scale methane emission from a boreal fen in relation to temperature, water table position, and carbon dioxide fluxes. *Global Biogeochemical Cycles*, 32, 1087–1106. https://doi.org/10.1029/2017GB005747

Sulman, B.N., Desai, A., Cook, B.D., Saliendra, N.Z., and Mackay, D.S. (2009). Contrasting carbon 820 dioxide fluxes between a drying shrub wetland in northern Wisconsin, USA, and nearby forests. 821 *Biogeosciences*. 6, 1115–1126. doi:10.5194/bg-6-1115-2009

Turner, J., Desai, A. R., Thom, J., Wickland, K. P., & Olson, B. (2019). Wind sheltering impacts on land-atmosphere fluxes over fens. *Frontiers in Environmental Science*, *7*, 179. https://doi.org/10.3389/fenvs.2019.00179

turner-j. (2021). turner-j/Methane-Flux-Plots: Methane Flux Lag Response Analysis (Version v0.1). Zenodo. http://doi.org/10.5281/zenodo.4617037

United States Geological Survey (USGS). (2020a). National Water Information System, "USGS 05357205 ALLEQUASH CREEK NEAR SAYNER, WI" [Data file]. Retrieved from https://maps.waterdata.usgs.gov/mapper

Updegraff, K., Bridgham, S.D., Pastor, J., Weishampel, P., Harth, C. (2001). Response of CO<sub>2</sub> and CH<sub>4</sub> emissions in peatlands to warming and water- table manipulation. *Ecological Applications*, **11**, 311–326. https://doi.org/10.1890/1051-0761(2001)011[0311:ROCACE]2.0.CO;2

Villa, J. A., Ju, Y., Vines, C., Rey- Sanchez, C., Morin, T. H., Wrighton, K. C., & Bohrer, G. (2019). Relationships between methane and carbon dioxide fluxes in a temperate cattail-

dominated freshwater wetland. *Journal of Geophysical Research: Biogeosciences*, *124*(7), 2076-2089. https://doi.org/10.1029/2019JG005167

Wei, G. W., Chen, Y., Sun, X. S., Chen, Y. H., Luo, F. L., & Yu, F. H. (2019). Growth responses of eight wetland species to water level fluctuation with different ranges and frequencies. *PloS one*, *14*(7), e0220231.

https://doi.org/10.1371/journal.pone.0220231

Welsch, M. and Yavitt, J.B. (2007), Microbial CO<sub>2</sub> production, CH<sub>4</sub> dynamics and nitrogen in a wetland soil (New York State, USA) associated with three plant species (*Typha, Lythrum, Phalaris*). European Journal of Soil Science, 58: 1493-1505. https://doi.org/10.1111/j.1365-2389.2007.00955.x

Zacks, G., Greet, J., Walsh, C. J., & Raulings, E. (2019). The flooding tolerance of two critical habitat-forming wetland shrubs, Leptospermum lanigerum and Melaleuca squarrosa, at different life history stages. *Australian Journal of Botany*, *66*(7), 500-510. https://doi.org/10.1071/BT18039

Zhang, Z., Zimmermann, N.E., Stenke, A., Li, X., Hodson, E.L., Zhu, G., Huang, C., and Poulter, B. (2017). Emerging role of wetland methane emissions in driving 21st century climate change. *Proceedings of the National Academy of Sciences*. 201618765; DOI:10.1073/pnas.1618765114 on CH4 and CO2 fluxes from peatlands converted to agriculture. *Science of the Total Environment*, 584, 665-672. https://doi.org/10.1016/j.scitotenv.2017.01.094

Nicolini, G., Castaldi, S., Fratini, G., & Valentini, R. (2013). A literature overview of micrometeorological CH4 and N2O flux measurements in terrestrial ecosystems. *Atmospheric Environment*, *81*, 311-319. https://doi.org/10.1016/j.atmosenv.2013.09.030

Nouchi, I., Mariko, S., & Aoki, K. (1990). Mechanism of Methane Transport from the Rhizosphere to the Atmosphere through Rice Plants. *Plant physiology*, *94*(1), 59–66. https://doi.org/10.1104/pp.94.1.59

Oikawa, P. Y., Jenerette, G. D., Knox, S. H., Sturtevant, C., Verfaillie, J., Dronova, I., ... & Baldocchi, D. D. (2017). Evaluation of a hierarchy of models reveals importance of substrate limitation for predicting carbon dioxide and methane exchange in restored wetlands. *Journal of Geophysical Research: Biogeosciences*, *122*(1), 145-167. https://doi.org/10.1002/2016JG003438

Olde, L. (2017). *How Do Rates of Carbon Metabolism Vary over a Geological Gradient, and How does this Contribute to Riverine Greenhouse Gas Emissions?* (Doctoral dissertation, Queen Mary University of London). https://qmro.qmul.ac.uk/xmlui/bitstream/handle/123456789/25933/Olde\_L\_PhD\_Final\_310717.

Olefeldt, D., Euskirchen, E. S., Harden, J., Kane, E., McGuire, A. D., Waldrop, M. P., & Turetsky, M. R. (2017). A decade of boreal rich fen greenhouse gas fluxes in response to natural and experimental water table variability. *Global change biology*, *23*(6), 2428-2440. https://doi.org/10.1111/gcb.13612

This article is protected by copyright. All rights reserved.

pdf?sequence=1&isAllowed=y